Author’s response to reviews

Title: A case of pleural effusion caused by Mycobacterium fortuitum and Mycobacterium mageritense coinfection

Authors:

Ryosuke Hirabayashi (rhirabayashi0405@kcho.jp)
Atsushi Nakagawa (nakagawa.atsushi.p40@kyoto-u.jp)
Hiroshi Takegawa (sbb-hiro@kcho.jp)
Keisuke Tomii (ktomii@kcho.jp)

Version: 2 Date: 03 Aug 2019

Author’s response to reviews:

10 August 2019

Oliver Cocks M.Sc.
Editor-in-chief
BMC Infectious Diseases

Dear Editor

Thank you for inviting us to submit a revised draft of our manuscript entitled, “A case of pleural effusion caused by Mycobacterium fortuitum and Mycobacterium mageritense coinfection” to BMC Infectious Diseases. We also appreciate the time and effort you and each of the reviewers have dedicated to providing insightful feedback on ways to strengthen our paper again.
Thus, it is with great pleasure that we resubmit our article for further consideration. We have incorporated changes that reflect the detailed suggestions you have graciously provided. We also hope that our edits and the responses we provide below satisfactorily address all the issues and concerns you and the reviewers have noted.

Sincerely

Ryosuke Hirabayashi
Department of Respiratory Medicine, Kobe City Medical Center General Hospital,
2-1-1 Minatojima-minamimachi, Chuo-ku, Kobe, Hyogo, 650-0047, Japan
Tel: +81-78-302-4321
Fax: +81-78-302-7537
Email: rhirabayashi0405@kcho.jp